Author’s response to reviews


Authors:

Simon JURIC (sj@inova.si)
Borut ZALIK (borut.zalik@um.si)

Version: 2 Date: 7 July 2014

Author’s response to reviews:

Hi,

1. Line Numbering

Has been added and the page breaks were removed.

2. Figures Title

Figure titles, legends have been edited to comply with the requested format - http://www.biomedcentral.com/authors/figures.

3. Figures

Have been removed from the document. We would just like to add that the small legend in Figure 6 (white rectangle box in the right bottom corner was a part of the acquired image via the ultrasound device and was not edited on our side, therefore that box still remains in the image). If necessary we can manually remove it, but we have added the legend regarding these values in the figure 6 information - in the main manuscript.

Thank you for your prompt feedback, please let us know if there are any additional changes needed and thank you for considering this manuscript.

Best Regards,
Simon